**Author's response to reviews**

**Title:** Full-term abdominal extrauterine pregnancy complicated by post-operative ascites with successful outcome: a case report

**Authors:**

Gwinyai Masukume (parturitions@gmail.com)
Elton Sengurayi (esengus@yahoo.com)
Alfred Muchara (atm4jc@gmail.com)
Emmanuel Mucheni (emucheni@yahoo.com)
Wedu Ndebele (wndebele@yahoo.co.uk)
Solwayo Ngwenya (drsolngwe@yahoo.co.uk)

**Version:** 4  **Date:** 4 August 2012

**Author’s response to reviews:**

Dear JMCR,

We state that intravenous fluids were administered as part of this patient’s treatment and also that she was informed about future care (documented on consent form). We feel it is important to state these points in the manuscript.

Sincerely yours,
GM, ES, AM, EN, WN & SN